Author's response to reviews

Title: Serum lipid responses to psyllium fiber: differences between pre- and post-menopausal, hypercholesterolemic women

Authors:

Vijay Ganji (vganji@gsu.edu)
Jennifer Kuo (jennkuo620@yahoo.com)

Version: 5 Date: 27 July 2008

Author's response to reviews:

Dear Editor-In-Chief:

Enclosed is the revised manuscript with format-related revisions. Please let me know if these revisions are acceptable. Thanks.

Vijay Ganji